communications medicine

ARTICLE

https://doi.org/10.1038/s43856-023-00353-2

OPEN

Waist-hip ratio is superior to BMI in predicting liver-related outcomes and synergizes with harmful alcohol use

Fredrik Åberg₁[™], Martti Färkkilä², Veikko Salomaa₁[®]³, Antti Jula³, Satu Männistö³, Markus Perola³, Annamari Lundqvist³ & Ville Männistö⁴

Abstract

Background: Obesity is associated with liver disease, but the best obesity-related predictor remains undefined. Controversy exists regarding possible synergism between obesity and alcohol use for liver-related outcomes (LRO). We assessed the predictive performance for LROs, and synergism with alcohol use, of abdominal obesity (waist-hip ratio, WHR), and compared it to overall obesity (body mass index, BMI).

Methods: Forty-thousand nine-hundred twenty-two adults attending the Finnish healthexamination surveys, FINRISK 1992-2012 and Health 2000 studies, were followed through linkage with electronic healthcare registries for LROs (hospitalizations, cancers, and deaths). Predictive performance of obesity measures (WHR, waist circumference [WC], and BMI) were assessed by Fine-Gray models and time-dependent area-under-the-curve (AUC).

Results: There are 355 LROs during a median follow-up of 12.9 years (509047.8 personyears). WHR and WC emerge as more powerful predictors of LROs than BMI. WHR shows significantly better 10-year AUC values for LROs (0.714, 95% CI 0.685-0.743) than WC (0.648, 95% CI 0.617-0.679) or BMI (0.550, 95% CI 0.514-0.585) both overall and separately among men and women. WHR is predictive also in BMI strata. Absolute 10-year risks of LROs are more dependent on WHR than BMI. Moreover, WHR shows a significant supra-additive interaction effect with harmful alcohol use for liver-related outcomes (excess 10-year cumulative incidence of 2.8% from the interaction), which is not seen between BMI and harmful alcohol use.

Conclusions: WHR is a better predictor than BMI or WC for LROs, and WHR better reflects the synergism with harmful alcohol use. WHR should be included in clinical assessment when evaluating obesity-related risks for liver outcomes.

Plain language summary

Obesity has been linked to liver disease, but the most accurate measure for predicting obesity-related liver disease outcomes remains uncertain. In this study, we analyzed data from over 40,000 adults to compare the extent to which different measures of obesity can predict liver-related outcomes, such as severe liver disease, liver failure, or death from liver disease. The measures of obesity were the ratio of waist circumference to hip circumference (waist-hip ratio, WHR), waist circumference (WC), and body mass index (BMI). Our findings reveal that WHR and WC are stronger predictors of these outcomes than BMI. In particular, WHR demonstrated superior predictive ability and this predictive ability was influenced by harmful alcohol use. This study suggests that WHR may be a relatively simple but useful measure for clinicians to use when predicting obesity-related risks for liver health.

¹Transplantation and Liver Surgery, Helsinki University Hospital and University of Helsinki, Helsinki, Finland. ²Abdominal Center, Helsinki University and Helsinki University Hospital, Helsinki, Finland. ³Finnish Institute for Health and Welfare, Helsinki, Finland. ⁴Departments of Medicine, University of Eastern Finland and Kuopio University Hospital, Kuopio, Finland. ^{Kel}email: Fredrik.Aberg@helsinki.fi

Check for updates

here is a dose-dependent relationship between the level of obesity and the risk for liver disease¹. However, the anthropometric measure that best predicts future liver disease remains unclear. While most studies have analyzed body mass index (BMI), it is now increasingly appreciated that the waist-hip ratio (WHR) better reflects fat distribution and abdominal obesity and as such seems to reflect metabolic health better than the BMI². Hip circumference mirrors lower-body subcutaneous fat mass, which is not as metabolically harmful as visceral fat mass in the abdominal region, and may even have protective effects^{2,3}. Waist and hip circumference have independent and opposite associations with incident liver disease, and these effects seem to be largely captured in the WHR⁴. The WHR has been shown to be the obesity indicator with the highest predictive capacity for non-alcoholic fatty liver disease (NAFLD)⁵. In addition, the WHR has been shown to predict severe liver disease and liver-related outcomes better than BMI, but population studies are still scarce⁶⁻⁸, and previous studies have not used competing-risk analysis.

In addition to obesity, there are also dose-dependent relationships between alcohol consumption and the risk for liver disease⁹. Moreover, many studies have pointed to supra-additive interaction effects between alcohol and obesity for markers of liver disease and for liver-related outcomes^{10,11}. Supra-additive interaction basically means that the combined risk effect of two concurrent exposures (for instance, harmful alcohol use and obesity) on the risk for liver disease is greater than the sum of their individual risk effects. Nonetheless, controversy remains regarding whether such interaction effects truly exist or not¹¹⁻¹³. A recent systematic review failed to find supra-additive interaction effects between alcohol and BMI for liver disease¹². However, there are several methodological concerns with the previous studies, including the reliance on self-reported data, often small sample sizes, and a lack of competing risk analyses and absolute risk estimates.

We assessed the predictive performance for liver-related outcomes of WHR, and compared it to that of waist circumference (WC) and BMI. We further compared the interaction effects between harmful alcohol use and WHR or BMI. Finally, we demonstrate how WHR affects the absolute risk of liver-related outcomes substantially more than the BMI when other parameters are kept constant. WHR emerged as a better predictor than BMI or WC for liver-related outcomes, and WHR better reflects the synergism with harmful alcohol use.

Methods

Data were sourced from the Finnish health-examination studies, FINRISK 1992–2012, and Health 2000 Survey. FINRISK studies are national population surveys carried out in Finland every 5 years by the Finnish Institute for Health and Welfare, using random representative population samples¹⁴. The Health 2000 Survey was similarly coordinated by the Finnish Institute for Health and Welfare, and originally comprised 8028 adults aged 30 years and above; the participation rate in the full examinations was 80%¹⁵.

Data collection, sample formation, and linkage with electronic healthcare registers for liver-related outcomes have been previously described^{4,14,15}. Briefly, weight, height, waist, and hip circumference were all measured at baseline, i.e., at the time of the health-examination. Alcohol use was assessed by standard questionnaires. Individuals were linked with the Care Register for Health Care (HILMO) for hospitalizations, with the Finnish Cancer Registry for malignancies, and with the Statistics Finland register for vital status and cause of death until December 2015. A liver-related outcome was defined by ICD-9 and ICD-10 codes reflecting severe liver disease (requiring hospital admission or causing liver cancer, or liver-related death) in line with a recent consensus paper¹⁶; the specific ICD codes used are presented in Supplementary Note 1.

All participants provided signed informed consent, and the studies were approved by the Coordinating Ethical Committee of the Helsinki and Uusimaa Hospital District. Previously, the studies were also approved by the institutional review board of the National Public Health Institute in Helsinki, Finland. The FINRISK 1992–2012 Health 2000 sample collections were transferred to THL Biobank in 2015 according to the Finnish Biobank Act.

Statistical methods. For comparing baseline characteristics between sex groups, we used Chi-Square or Mann-Whitney tests as appropriate. Correlations were calculated using the Spearman method. Associations between WHR, WC, or BMI with liver-related outcomes were assessed by Fine-Gray regression analyses, and non-linear associations by restricted cubic splines. We evaluated the discrimination performance of WHR, WC, and BMI for liver-related outcomes in terms of time-dependent area-under-the-curve (AUC) values at 10 years of follow-up based on Fine-Gray competing-risk regression models, where death without liver disease was considered a competing-risk event. Models were compared statistically by delta-AUCs and the Wald test using the methodology described by Blanche et al.¹⁷. We also evaluated the discrimination performance of WHR and WC in BMI strata to see if the performance of these obesity measures depended on the BMI.

To illustrate how the absolute risk of liver-related outcomes vary by WHR and BMI when age, sex, and alcohol use are kept constant, we constructed a Fine-Gray competing-risk model, separately for men and women, with age, alcohol use, WHR, and BMI as independent variables. Based on this model, we predicted the 10-year absolute risk for liver-related outcomes for a set of pre-specified parameters (BMI 25 or 35 kg/m², and WHR 1.2 or 0.9 for men or 1.0 or 0.7 for women). We repeated this procedure in the subgroup of individuals with a high risk of having advanced liver fibrosis at baseline. Here, fibrosis risk was determined using the dynamic aspartate-to-alanine aminotransferase ratio (dAAR score), which is an externally validated score that is associated with both advanced fibrosis/cirrhosis and liver-related outcomes¹⁸. High risk was defined as a dAAR score above 2.63, in line with the original publication¹⁸.

To analyze whether supra-additive interaction effects exist between WHR or BMI and harmful alcohol use for liver-related outcomes, we applied the methodology recently described by Innes et al.¹³. First, we stratified WHR, BMI, and alcohol use into three levels. WHR was stratified into sex-specific tertiles (cutoffs for men: 0.93 and 0.99; cutoffs for women: 0.80 and 0.86) because there are no firmly established WHR cutoffs. BMI was stratified into normal weight (BMI 20-24.9 kg/m²), overweight (BMI 25-29.9 kg/m²), and obese (BMI \geq 30 kg/m²) categories, and alcohol use according to the UK guidelines for safe, hazardous, and harmful drinking by the following cutoffs (men: 176 and 392 grams of ethanol/week; women: 120 and 280 grams/week)¹⁹. The reference group was those with safe alcohol use and no overweight or obesity (BMI 20-24.9 kg/m² or lowest WHR tertile). Then, using the nonparametric cumulative incidence function, we calculated the 10-year excess cumulative incidence of liver-related outcomes by subtracting the cumulative incidence observed in a specific group from the cumulative incidence in the reference group. We repeated this procedure to calculate relative risks by Fine-Gray regression models adjusted for age, sex, education level, and employment and marital status. A two-tailed P-value < 0.05 was considered statistically significant. Data were analyzed with R software version 3.6.1.

Reporting summary. Further information on research design is available in the Nature Portfolio Reporting Summary linked to this article.

Results

Study population. The initial combined sample from all surveys comprised 43,105 individuals. After excluding individuals with missing registry linkage (n = 1457), baseline liver disease (ICD-10 codes K70-K77 or C22; n = 299), chronic viral hepatitis (n = 89), or missing anthropometric measurements (n = 338), the final study cohort comprised 40,922 individuals.

Baseline characteristics of the 40,922 study participants are shown in Table 1. Mean age was 50 years, 47% were men, and the mean BMI was 26.8 kg/m². Mean WHR was 0.96 among men and 0.84 among women. Mean WC was 96.3 cm among men 84.9 cm

among women. Mean alcohol use was 75 grams of ethanol per week (around 7.5 standard units per week), 115 g/week for men and 39 g/week for women. Of the participants, 86% were in the lowest alcohol use strata (low alcohol use), 10% in the middle, and 4% in the highest strata (harmful alcohol use). For WHR, the distribution from the lowest to highest strata were 31%, 38%, and 31% respectively, and for BMI, 35%, 40%, and 21%, respectively; another 4% had a BMI < 20 kg/m².

Associations between baseline variables. WHR correlated with BMI among both men (r = 0.64, 95% CI 0.63–0.64, P < 0.001) and women (r = 0.54, 95% CI 0.53–0.55, P < 0.001) (Fig. 1). None-theless, 2128 (37%) of men and 4537 (53%) of women with a BMI 20–25 kg/m² still had a WHR in the middle or highest sex-specific tertile. On the other hand, among men with a BMI above

	Overall	Men	Women	Р
Participants	40,922	19,399	21,523	
Age, years	49.6 (13.8)	49.9 (13.6)	49.3 (14.0)	< 0.001
Education				0.001
Low	13,403 (33.2)	6179 (32.3)	7224 (34.0)	
Average	13,243 (32.8)	6419 (33.6)	6824 (32.1)	
High	13,701 (34.0)	6508 (34.1)	7193 (33.9)	
Marital status				< 0.001
Married/partnership	29,654 (72,6)	14,737 (76,1)	14,917 (69,5)	
Single	5416 (13.3)	2835 (14.6)	2581 (12.0)	
Widow, separated, divorced	5764 (14.1)	1784 (9.2)	3980 (18.5)	
Employment status				< 0.001
Part- or full-time employed	24,788 (61,4)	12.051 (63.1)	12,737 (59.8)	
Other ^a	5083 (12.6)	1801 (9.4)	3282 (15.4)	
Retired	10 526 (26 1)	5261 (27 5)	5265 (247)	
Diabetes	3097 (7.6)	1555 (8.0)	1542 (7.2)	0.001
Waist circumference cm	90.3 (13.8)	96.3 (11.8)	84.9 (13.3)	<0.001
Hip circumference cm	100 9 (91)	100 4 (7 5)	101 4 (10 3)	<0.001
Waist-hin ratio	0.89 (0.09)	0.96 (0.07)	0.84 (0.07)	<0.001
Waist-hip ratio category				< 0.001
Lowest	12,703 (31,0)	6276 (32.4)	6427 (29.9)	
Middle	15 368 (37 6)	7290 (37.6)	8078 (375)	
Highest	12 851 (31 4)	5833 (30.1)	7018 (32.6)	
Body mass index kg/m^2	26.8 (4.7)	271(41)	265 (51)	<0.001
Body mass index, kg/m Rody mass index category kg/m ²	20.0 (1.7)	27.11 (1.17)	20.0 (0.1)	<0.001
<20	1434 (35)	288 (15)	1146 (53)	-0.001
20-24.9	14 353 (35 2)	5813 (30.1)	8540 (39.8)	
25-29.9	16 297 (40.0)	9172 (47.5)	7125 (33.2)	
>30	8680 (213)	1025 (20.9)	/655 (217)	
Weekly alcohol use g	74 9 (136 6)	115 0 (172 3)	389 (77 /)	<0.001
	74.7 (150.0)	115.0 (172.5)	56.7 (77.4)	<0.001
	33 535 (86 3)	14 645 (79 6)	18 800 (02 3)	SU.001
Middle	39/2 (10.1)	2688 (14.6)	10,070 (72.3)	
Highest	1202 (2 4)	1064 (59)	210 (1 4)	



Fig. 1 Correlations between waist-hip ratio and body mass index. Scatter plots showing the correlation between waist-hip ratio and body mass index in men (a) and women (b). Vertical lines represent the sex-specific tertile cutoffs. Horizontal lines separate normal weight from overweight and obesity.

30 kg/m², 1211 (30%) had a WHR in the middle or lowest sexspecific tertile. Likewise, 1545 (33%) of women with BMI > 30 kg/m² had a WHR in the middle or lowest tertile. The correlation between WHR and WC was 0.79 (95% CI 0.78–0.79, P < 0.001) for men and 0.81 (95% CI 0.80–0.81, P < 0.001) for women. Correlation coefficients between WHR and weekly alcohol use were 0.04 (P < 0.001) for men and -0.08 (P < 0.001) for women.

Liver-related outcomes. We observed 355 liver-related outcome events during a median follow-up of 12.9 years (IQR 7.8–17.8, 509,047.8 person-years). Of the liver events, 198 (56%) were initially coded as alcohol-related liver disease, 109 (31%) as other chronic liver disease, and 48 (13%) as hepatocellular carcinoma. There were respectively 185, 83, and 71 liver-related events in the lowest, middle, and highest alcohol use strata. The respective figures for the WHR strata were 84, 98, and 173, and for the BMI strata, 87, 139, and 116. In addition, 11 liver events occurred in those with BMI < 20 kg/m².

WHR is a better predictor of liver-related outcomes than BMI. By univariate Fine-Gray regression analysis accounting for nonlinear associations, there was no evidence that the association between WHR and the rate of liver-related outcomes was non-linear (P for non-linearity, 0.22); the same was true for WC and liver outcomes (P for non-linearity, 0.44) (Fig. 2). Regarding BMI, the association was U-shaped (P for non-linearity, 0.01), meaning that both a low and a high BMI increased the rate of liver-related outcomes, compared to a BMI around 20-25 kg/m². In addition, WHR and WC emerged as more powerful predictors of liver-related outcomes than BMI, since the hazard ratios per 1 SD were 1.99 (95% CI 1.83-2.15) for WHR, 1.79 (95% CI 1.63-1.97) for WC, but only 1.35 (95% CI 1.23-1.48) for BMI. Even after the removal of those with $BMI < 20 \text{ kg/m}^2$ (i.e., the subgroup that caused the non-linear association between BMI and liver-related outcomes), the hazard ratio per 1 SD was 1.37 (95% CI 1.25-1.50) for BMI.

The discrimination performance (10-year AUC values) of WHR for liver-related outcomes was superior to that of BMI and WC, overall and separately among men and women, although the 95% CIs became wide in sex-specific analyses (Table 2).

In the subgroup of safe and current alcohol users (n = 27,781 with 149 liver events), 10-year AUC values for WHR, BMI, and WC were 0.70 (95% CI 0.64–0.76), 0.61 (95% CI 0.54–0.68), and 0.67 (95% CI 0.60–0.73), respectively. In the subgroup of lifetime alcohol abstainers (n = 3587 with 16 liver events), the corresponding AUC values were 0.67 (95% CI 0.46–0.88), 0.64 (95% CI 0.40–0.88), and 0.68 (0.47–0.88), respectively.

Effect modification and combined associations of abdominal obesity and BMI for liver-related outcomes. When examined in BMI strata, the superiority of WHR over WC in terms of discrimination was most clear in individuals with normal weight or overweight according to the BMI (Table 3).

Figure 3 shows how the absolute risks of liver-related outcomes change, for example, individuals when age and alcohol use are kept constant, but WHR and BMI vary. As demonstrated, the absolute risks are more dependent on WHR than BMI. Furthermore, when the same analysis was repeated in the subgroup of 389 individuals with elevated baseline dAAR scores (i.e., high risk of having advanced liver fibrosis at baseline), the absolute risk estimates for liver-related outcomes increased



Fig. 2 Anthropometric measures and liver-related outcomes. Influences on risk for liver-related outcomes of body mass index (a), waist-hip ratio (b), and waist circumference (c) by Fine-Gray regression analysis accounting for non-linear associations. Shaded areas represent 95% confidence intervals.

	All		Men		Women	
	10-year AUC (95% CI)	Pa	10-year AUC (95% CI)	Pa	10-year AUC (95% CI)	Pa
Waist-hip ratio	0.714 (0.685-0.743)		0.697 (0.489-0.905)		0.679 (0.464-0.895)	
Waist circumference	0.648 (0.617-0.679)	<0.001	0.640 (0.445-0.835)	0.002	0.611 (0.414-0.808)	0.02
Body mass index (linear)	0.550 (0.514-0.585)	<0.001	0.568 (0.392-0.745)	<0.001	0.568 (0.380-0.757)	0.02
Body mass index (non-linear)	0.598 (0.557-0.640)	<0.001	0.619 (0.570-0.667)	0.006	0.573 (0.495-0.651)	0.02

Table 2 Area-under-the-curve (AUC) values at 10 years for the various anthropometric measures in predicting liver-related outcomes.

substantially, highlighting how the effects of WHR becomes even more relevant when the background risk increases (Fig. 3).

Supra-additive interaction between WHR and harmful alcohol use for liver-related outcomes. Next, we assessed the excess cumulative incidence of severe liver disease after 10 years of follow-up according to harmful alcohol use and the highest risk category of WHR or BMI. When compared to the incidence of the reference group with low alcohol use and the lowest sexspecific WHR tertile (no abdominal obesity), those with harmful alcohol use and in the lowest WHR tertile (no abdominal obesity) had an excess cumulative incidence of liver-related outcomes of 0.88% at 10 years (Fig. 4). Those in the highest sex-specific WHR tertile (abdominal obesity) with low alcohol use had an excess incidence of 0.28%. Finally, among those in the highest sexspecific WHR tertile (abdominal obesity) and with harmful alcohol use, the excess cumulative incidence was 3.96% at 10 years, resulting in a supra-additive interaction effect equal to 2.80% from the combination of abdominal obesity and harmful alcohol use (Fig. 4). With regard to BMI, a similar interaction effect with harmful alcohol use was minimal (0.2%) and nonsignificant (Fig. 4). Similar supra-additive interaction effects between WHR and harmful alcohol use, but not between BMI and harmful alcohol use, were confirmed in multivariableadjusted Fine-Gray regression analyses (Fig. 4).

When using the middle sex-specific WHR strata instead of the highest strata to define abdominal obesity, a supra-additive interaction effect equal to 2.77% at 10 years from the combination of abdominal obesity and harmful alcohol use remained. When considering the middle strata of BMI, the interaction effect was negative (-0.97% at 10 years).

Discussion

We found that WHR predicted liver-related outcomes in the general population better than BMI or WC. Moreover, the performance of WHR was preserved in various BMI strata. We further demonstrated that the absolute risk of liver-related outcomes is substantially more dependent on the WHR than on the BMI, when age, sex, and alcohol use are kept constant. The superiority of WHR over the other anthropometric measures for predictions of liver-related outcomes has previously been demonstrated in two studies^{6,7}. This evidence suggests that the WHR should become the standard obesity measure when assessing risk for liver-related outcomes. We acknowledge, however, that this might differ by ethnicity, which calls for further study in diverse ethnic groups. Nonetheless, any lack of WHR data in many datasets and clinical databases should not impede the introduction of the WHR in future studies and clinical practice.

Besides better predictive performance, WHR showed a significant supra-additive interaction effect with harmful alcohol use for liver-related outcomes, which was not seen between BMI and harmful alcohol use. This might explain the mixed findings regarding interaction effects between obesity and alcohol use in previous studies, most of which have assessed obesity using BMI^{10–12}. WHR might better reflect metabolic health than BMI^{2,3}. Metabolic dysfunction might sensitize the liver to the harmful effects of alcohol, as suggested by animal studies^{20,21} and human experimental studies²² and substantiated by recent epidemiologic evidence^{23–25}.

The WHR is incorporated in the recent Chronic Liver Disease (CLivD) risk score, which can be used to predict incident liverrelated outcomes in the general population²⁶. The CLivD score simultaneously considers both alcohol use, abdominal obesity (WHR), age, sex, diabetes, smoking, and gamma-glutamyltransferase. The CLivD score has been validated in external cohorts^{26,27}.

BMI showed a *U*-shaped association with liver-related outcomes, which has also been reported previously²⁸. Based on this, it seems that losing weight is not necessarily always a good thing if this weight loss means that the individual is losing muscle mass, for example, due to illness. In contrast, losing weight in a way that reduces the WC in relation to the hip circumference emerges as a better marker of reduction of liver risk²⁹.

It is noteworthy that, even at normal BMI levels $(20-25 \text{ kg/m}^2)$, a substantial proportion of men and women had elevated WHRs. This calls to question the appropriateness of using BMI to define a lean type of NAFLD. The WHR might more accurately separate abdominally obese (metabolically unhealthy) NAFLD from non-abdominally obese NAFLD.

We found that body fat distribution reflected by the WHR was a stronger determinant of absolute risks for liver-related outcomes than BMI. Although the absolute risks remained low in our example individuals shown in Fig. 3 in the general population overall, we found that the absolute risks increased substantially in the subgroup of individuals with a high risk of having advanced liver fibrosis/cirrhosis at baseline. This supports the need for assessment of liver fibrosis as part of risk stratification strategies.

The WHR showed superior performance over WC in our study. Based on the same dataset, Danielsson et al. previously found that hip circumference has independent predictive value for liver-related outcomes⁴. However, the added predictive information from hip circumference is largely captured in the WHR, even though the WHR does not reflect the absolute values of WC or hip circumference, only their ratio⁴.

Strengths of our study include the large dataset representative of the general population and with long-term follow-up for clinical liver-related outcomes. To our knowledge, this is the first study to assess the predictive performance of different anthropometric measures for liver-related outcomes using competingrisk analysis.

In this study, WHR was measured using a standardized protocol. However, the validity of self-measurements has been

Body mass index, kg/m ²	n/N	10-year AUC				
		Waist-hip ratio	Waist circumference	Pa		
<20	11/1433	0.871 (0.737-1.000)	0.812 (0.692-0.933)	0.20		
20-24	87/14353	0.749 (0.567-0.931)	0.660 (0.494-0.826)	< 0.00		
25-29	139/16297	0.711 (0.504-0.918)	0.668 (0.473-0.864)	0.02		
30-34	84/6364	0.730 (0.567-0.894)	0.723 (0.558-0.888)	0.70		
≥35	32/2316	0.687 (0.539-0.835)	0.741 (0.597-0.886)	0.20		

^aComparison between waist-hip ratio and waist circumference.



Fig. 3 Estimated risks of liver-related outcomes according to the anthropometric profile. The effects of waist-hip ratio (WHR) and body mass index (BMI) on the absolute 10-year risk of liver-related outcomes (%) in men and women when age and alcohol use are kept constant. Risks are shown for the population overall and separately for those with high risk of having advanced liver fibrosis at baseline according to the dynamic aspartate aminotransferaseto-alanine aminotransferase ratio (dAAR) score. Analyses are by Fine-Gray competing-risk regression.



Fig. 4 Interaction between harmful alcohol use and anthropometric measures. Excess cumulative incidence at 10 years of liver-related outcomes according to harmful alcohol use and waist-hip ratio (WHR) or body mass index (BMI).

confirmed previously^{30,31}, and self-measurements could help make WHR more attainable in clinical practice. Recent mobile applications enable valid WHR measurements using digital photography technology³², and the introduction of such applications to the clinic is awaited to further increase the implementation of the WHR.

Study limitations include the relatively low number of liverrelated outcome events and the reliance on registry-based outcomes. The Finnish population is predominantly white people, which might have an effect on results. Nonetheless, WHR has shown superiority over BMI also, for instance, in some previous Chinese studies^{5,33}.

In conclusion, WHR predicted liver-related outcomes better than BMI or WC in our large cohort representative of the general population. We found supra-additive interaction effects between WHR and harmful alcohol use for liver-related outcomes, which were not seen between BMI and harmful alcohol use. In white people of European descent, the WHR merits to become the standard obesity measure when assessing risk for liver-related outcomes. Therefore, WHR needs to be implemented in clinical practice.

Data availability

FINRISK and Health 2000 data are available from the THL biobank based on a research application, as explained on the website of the THL biobank (https://thl.fi/en/web/thl-biobank/for-researchers). Source data for the figures contain individual-level data that cannot be made publicly available, but these data can be retrieved through research applications to the THL biobank.

Received: 5 February 2023; Accepted: 31 August 2023; Published online: 06 September 2023

References

- Hagström, H., Tynelius, P. & Rasmussen, F. High BMI in late adolescence predicts future severe liver disease and hepatocellular carcinoma: a national, population-based cohort study in 1.2 million men. *Gut.* 67, 1536–1542 (2018).
- Stefan, N. Causes, consequences, and treatment of metabolically unhealthy fat distribution. *Lancet Diabetes Endocrinol.* 8, 616–627 (2020).
- Lotta, L. A. et al. Integrative genomic analysis implicates limited peripheral adipose storage capacity in the pathogenesis of human insulin resistance. *Nat. Genet.* 49, 17–26 (2017).
- Danielsson, O. et al. Waist and hip circumference are independently associated with the risk of liver disease in population-based studies. *Liver Int.* 41, 2903–2913 (2021).
- Zheng, R.-D., Chen, Z.-R., Chen, J.-N., Lu, Y.-H. & Chen, J. Role of Body Mass Index, waist-to-height and waist-to-hip ratio in prediction of nonalcoholic fatty liver disease. *Gastroenterol. Res. Pract.* 2012, 362147 (2012).
- Andreasson, A., Carlsson, A. C., Önnerhag, K. & Hagström, H. Waist/hip ratio better predicts development of severe liver disease within 20 years than body mass index: a population-based Cohort Study. *Clin. Gastroenterol. Hepatol.* 15, 1294–1301.e2 (2017).
- Schult, A., Mehlig, K., Björkelund, C., Wallerstedt, S. & Kaczynski, J. Waist-tohip ratio but not body mass index predicts liver cirrhosis in women. *Scand. J. Gastroenterol.* 53, 212–217 (2018).
- Åberg, F. & Jula, A. The sagittal abdominal diameter: role in predicting severe liver disease in the general population. *Obes. Res. Clin. Pract.* 12, 394–396 (2018).
- Roerecke, M. et al. Alcohol consumption and risk of liver cirrhosis: a systematic review and meta-analysis. *Am. J. Gastroenterol.* 114, 1574–1586 (2019).
- Hart, C. L., Morrison, D. S., Batty, G. D., Mitchell, R. J. & Davey Smith, G. Effect of body mass index and alcohol consumption on liver disease: analysis of data from two prospective cohort studies. *BMJ*. 340, c1240 (2010).
- Åberg, F., Färkkilä, M. & Männistö, V. Interaction between alcohol use and metabolic risk factors for liver disease: a critical review of epidemiological studies. *Alcohol Clin. Exp. Res.* 44, 384–403 (2020).
- 12. Glyn-Owen, K., Böhning, D., Parkes, J., Roderick, P. & Buchanan, R. The combined effect of alcohol and body mass index on risk of chronic liver

disease: a systematic review and meta-analysis of cohort studies. *Liver Int.* **41**, 1216–1226 (2021).

- Innes, H. et al. Characterising the risk interplay between alcohol intake and body mass index on cirrhosis morbidity. *Hepatology* 75, 369–378 (2021).
- Borodulin, K. et al. Cohort profile: The National FINRISK Study. Int. J. Epidemiol. 47, 696–696i (2018).
- Aromaa, A. & Koskinen, S. Health and Functional Capacity in Finland: Baseline Results of the Health 2000 Health Examination Survey. (Publications of National Public Health Institute, Series B 12/2004, Helsinki, Finland, 2004). [Internet]. [cited 2021 Aug 26]; Available from: https://www.julkari.fi/handle/ 10024/78534.
- Hagström, H. et al. Administrative coding in electronic health care recordbased research of NAFLD: an expert panel consensus statement. *Hepatology* 74, 474–482 (2021).
- Blanche, P. et al. Quantifying and comparing dynamic predictive accuracy of joint models for longitudinal marker and time-to-event in presence of censoring and competing risks. *Biometrics* 71, 102–113 (2015).
- Åberg, F. et al. A dynamic aspartate-to-alanine aminotransferase ratio provides valid predictions of incident severe liver disease. *Hepatol. Commun.* 5, 1021–1035 (2021).
- NICE. Overview | Alcohol-Use Disorders: Prevention | Guidance | NICE [Internet]. [cited 2022 Oct 7]; Available from: https://www.nice.org.uk/ guidance/ph24 (NICE, 2010).
- Xu, J. et al. Synergistic steatohepatitis by moderate obesity and alcohol in mice despite increased adiponectin and p-AMPK. *J. Hepatol.* 55, 673–682 (2011).
- Minato, T. et al. Binge alcohol consumption aggravates oxidative stress and promotes pathogenesis of NASH from obesity-induced simple steatosis. *Mol. Med.* 20, 490–502 (2014).
- 22. Israelsen, M. et al. Comprehensive lipidomics reveals phenotypic differences in hepatic lipid turnover in ALD and NAFLD during alcohol intoxication. *JHEP Rep.* **3**, 100325 (2021).
- Younossi, Z. M. et al. Effects of alcohol consumption and metabolic syndrome on mortality in patients with nonalcoholic and alcohol-related fatty liver disease. *Clin. Gastroenterol. Hepatol.* 17, 1625–1633.e1 (2019).
- Åberg, F., Helenius-Hietala, J., Puukka, P. & Jula, A. Binge drinking and the risk of liver events: A population-based cohort study. *Liver Int.* 37, 1373–1381 (2017).
- Åberg, F., Byrne, C. D., Pirola, C. J., Männistö, V. & Sookoian, S. Alcohol consumption and metabolic syndrome: Clinical and epidemiological impact on liver disease. *J. Hepatol.* 78, 191–206 (2023).
- Åberg, F. et al. Development and validation of a model to predict incident chronic liver disease in the general population: the CLivD score. J. Hepatol. 77, 302–311 (2022).
- Song, J. & Jiang, Z. G. A good step toward low-cost prognostication of liverrelated outcome awaits more validation. J. Hepatol. 77, 887–889 (2022).
- Liu, B., Balkwill, A., Reeves, G. & Beral, V. Million Women Study Collaborators. Body mass index and risk of liver cirrhosis in middle aged UK women: prospective study. *BMJ* 340, c912 (2010).
- Åberg, F., Britton, A. & Luukkonen, P. K. Changes over time in the Chronic Liver Disease risk score predict liver-related outcomes: longitudinal analysis of the Whitehall II study. *Scand. J. Gastroenterol.* 58, 170–177 (2023).
- Roberts, C. A., Wilder, L. B., Jackson, R. T., Moy, T. F. & Becker, D. M. Accuracy of self-measurement of waist and hip circumference in men and women. J. Am. Diet Assoc. 97, 534–536 (1997).
- Barrios, P., Martin-Biggers, J., Quick, V. & Byrd-Bredbenner, C. Reliability and criterion validity of self-measured waist, hip, and neck circumferences. *BMC Med. Res. Methodol.* 16, 49 (2016).
- Neufeld, E. V., Seltzer, R. A., Sazzad, T. & Dolezal, B. A. A multidomain approach to assessing the convergent and concurrent validity of a mobile application when compared to conventional methods of determining body composition. *Sensors (Basel)*. 20, E6165 (2020).
- Pang, Y. et al. Central adiposity in relation to risk of liver cancer in Chinese adults: A prospective study of 0.5 million people. *Int. J. Cancer* 145, 1245–1253 (2019).

Acknowledgements

The samples/data used for the research were obtained from THL Biobank (study numbers: BB2016_98, BB2017_101, and BB2019_31). We thank all study participants for their generous participation at THL Biobank and the National FINRISK Study (years 1992, 1997, 2002, 2007, 2012) and the Health 2000 Survey. Financial support: Dr. Åberg was supported by Finska Läkaresällskapet, Academy of Finland (#338544), and Sigrid Jusélius Foundation. Dr. Salomaa was supported by the Finnish Foundation for Cardiovascular Research and by the Juho Vainio Foundation. Dr. Männistö was supported by the Finnish Medical Foundation and State Research Funding (VTR). The researchers are all independent of the funders.

ARTICLE

Author contributions

Study conception (F.Å., V.M., M.F.), data collection (V.S., A.J., S.M., M.P., A.L.), statistical analyses (F.Å.), interpretation of results (all authors), wrote the first draft (F.Å.), critical revision (all authors).

Competing interests

The authors declare no competing interests.

Additional information

Supplementary information The online version contains supplementary material available at https://doi.org/10.1038/s43856-023-00353-2.

Correspondence and requests for materials should be addressed to Fredrik Åberg.

Peer review information *Communications Medicine* thanks Yoji Ishizu and the other, anonymous, reviewer(s) for their contribution to the peer review of this work.

Reprints and permission information is available at http://www.nature.com/reprints

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/ licenses/by/4.0/.

© The Author(s) 2023